Author's response to reviews

Title: Risk Behaviours related to HIV among Taxi Drivers and Their Assistants in Addis Ababa, Ethiopia

Authors:

Yihunie Mr Lakew Mr (yihunierh@yahoo.com)
Habtamu Mr Tamene Mr (habtamu@psi.org.et)

Version: 3 Date: 7 March 2013

Author's response to reviews:

Pont by Point Responses
This article was submitted to the Reproductive Health Journal and the journal editor-in–chief, recommended it to be published in BMC Public Health. Currently, it is not being considered by any other journal for publication.
The Journal of Health Promotion Perspectives was mentioned by mistake in the previous cover letter send to you.
The manuscript is edited by a professional native language speaker. The revised article is attached.
Sincerely,
Yihunie Lakew (Mr)